Author’s response to reviews

Title: Fungaemia caused by obstructive renal candida bezoars leads to bilateral chorioretinitis: a case report

Authors:

Johannes Stein (johannes.stein@ukbonn.de)
Stefan Latz (stefan.latz@ukbonn.de)
Jörg Ellinger (joerg.ellinger@ukbonn.de)
Guido Fechner (guido.fechner@ukbonn.de)
Maher Safi (maher.safi@ukbonn.de)
Philipp Krausewitz (Philipp.Krausewitz@ukbonn.de)
Simone Müller (simone.mueller@ukbonn.de)
Karin Weyer (Karin.weyer@ukbonn.de)
Stefan C Müller (stefan.mueller@ukbonn.de)

Version: 3 Date: 18 Feb 2018

Author’s response to reviews:

Dear reviewers,

Thank you very much for the interesting suggestions.

Reviewer 1, Dr. Haymann:

"I think it would be interesting to replace figure 2 which shows the ureteroscopy view (which is blurred and doesn't bring many information), by the CT scan image showing left renal obstruction: presence of air (emphysematous pyelitis?)? density of the mass in Hounsfield units?"

As recommended we have replaced figure 2 by the CT scan (now figure 1), which shows the renal obstruction (Hounsfield units 13) caused by the bezoars. There was no presence of air.

"The fact that renal drainage was not performed at admission should be commented in the discussion"
We have added a section in the discussion whether an initial percutaneous nephrostomy or an internal DJ stent is preferable in these cases. (e.g. Discussion, Page 5, Line 9-13).

Best regards,

J. Stein